 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=50pt 1301pt 2087pt 50pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row1col1}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row2col1}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=295pt 1301pt 1842pt 50pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row1col2}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row2col2}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=539pt 1301pt 1598pt 50pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row1col3}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row2col3}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=784pt 1301pt 1353pt 50pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row1col4}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row2col4}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1028pt 1301pt 1109pt 50pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row1col5}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row2col5}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1273pt 1301pt 864pt 50pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row1col6}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row2col6}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1517pt 1301pt 620pt 50pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row1col7}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row2col7}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1762pt 1301pt 375pt 50pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row1col8}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row2col8}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=50pt 964pt 2087pt 387pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row2col1}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row3col1}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row1col1}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=295pt 964pt 1842pt 387pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row2col2}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row3col2}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row1col2}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=539pt 964pt 1598pt 387pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row2col3}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row3col3}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row1col3}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=784pt 964pt 1353pt 387pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row2col4}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row3col4}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row1col4}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1028pt 964pt 1109pt 387pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row2col5}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row3col5}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row1col5}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1273pt 964pt 864pt 387pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row2col6}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row3col6}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row1col6}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1517pt 964pt 620pt 387pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row2col7}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row3col7}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row1col7}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1762pt 964pt 375pt 387pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row2col8}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row3col8}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row1col8}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=50pt 627pt 2087pt 724pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row3col1}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row4col1}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row2col1}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=295pt 627pt 1842pt 724pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row3col2}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row4col2}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row2col2}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=539pt 627pt 1598pt 724pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row3col3}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row4col3}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row2col3}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=784pt 627pt 1353pt 724pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row3col4}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row4col4}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row2col4}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1028pt 627pt 1109pt 724pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row3col5}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row4col5}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row2col5}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1273pt 627pt 864pt 724pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row3col6}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row4col6}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row2col6}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1517pt 627pt 620pt 724pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row3col7}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row4col7}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row2col7}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1762pt 627pt 375pt 724pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row3col8}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{\pageref{row4col8}}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row2col8}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=50pt 290pt 2087pt 1061pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row4col1}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row3col1}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=295pt 290pt 1842pt 1061pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row4col2}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row3col2}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=539pt 290pt 1598pt 1061pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row4col3}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row3col3}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=784pt 290pt 1353pt 1061pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row4col4}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row3col4}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1028pt 290pt 1109pt 1061pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row4col5}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row3col5}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1273pt 290pt 864pt 1061pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row4col6}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row3col6}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1517pt 290pt 620pt 1061pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row4col7}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row3col7}}}%
\newpage


 
\resizebox{\paperwidth}{\paperheight}{%
\adjustbox{trim=1762pt 290pt 375pt 1061pt,clip}{\usebox{\mybox}}}%
\phantomsection\label{row4col8}%
\hspace*{-0.5\paperwidth}%
\vspace*{-4cm}%
\raisebox{1cm}[0in][0in]{\makebox[0em]{}}%
\raisebox{11cm}[0in][0in]{\makebox[0em]{\pageref{row3col8}}}%
\newpage


